Author’s response to reviews

Title: Changes in food intake patterns during 2000-2007 and 2008-2016 in the population-based prospective Northern Sweden Diet Database

Authors:

Ena Huseinovic (ena.huseinovic@gu.se)

Agneta Hörnell (agneta.hornell@umu.se)

Ingegerd Johansson (ingejerd.johansson@umu.se)

Anders Esberg (anders.esberg@odont.umu.se)

Bernt Lindahl (bernt.lindahl@umu.se)

Anna Winkvist (anna.winkvist@nutrition.gu.se)

Version: 3 Date: 28 Jun 2019

Author’s response to reviews:

Dear Sir,

Thanks for making me aware of a new version of the Global Burden of Disease project focusing on diet. I have now included this in the first paragraph of the Introduction instead of the older version, and also updated the reference list. These changes are indicated in red.